Notes of the RICH1 Phone meeting 27/04/2007
People Present:
Bristol: Richard

CERN: Carmelo(part), Dave, Fabio, Olav, Richard P., Werner
IC:  Bill, Trevor(part)      
Oxford: Brian
Apologies for absence:
Neville, Nick
Discussed Agenda Topics:
1. Spherical Mirrors

Mechanics: Waiting for the V-blocks to be machined at a compound angle at IC. These should be ready by Tuesday, and then all mechanics will be ready.
Installation trails: A number of small problems were found.  Machining was required to correct these minor problems. The A-side spherical mirror C-frame was installed in 156 and all went smoothly. The upper bridge rail is difficult to access when the C-frame is in place. The C-frame is 2-3 mm too high and may need to be adjusted. This will be confirmed at the Gas enclosure next week.

2. Flat Mirrors
Mechanics: Claws for the angle adjusters are ready and will be delivered to CERN next week. The remaining two V-blocks will be ready next week.
Installation trails: Several problems were found which were adjusted by machining. There have been problems in putting in the rails in the installation boxes because the welds were not flush. Tim ground the welds down. The joining rail between the boxes and rig was too short by 10 cm. Tim machined 2 new joining rails. The plates were not slid onto the rig because the adjusters and claws were not available. The plates with the side handles do not fit in the boxes.  The side-handles will need to be removed each time. The adjusters are ready and will be delivered next week. The boxes had the tendency to tip. Richard corrected this by fixing two side bars onto the boxes. The holding bracket hole on the installation plate needs extension (manufacturer error). Tim will do this next time he is at CERN. The lower H-frame needed modification (addition of a bottom bar) because the holes were in the wrong position. 

Tim and Richard were out at CERN in these last weeks. Their help was essential in testing the mechanics because several parts needed machining. Tim will be coming out again from the 7-11 May 07.
3. Mirror Alignment
The plan is to align first the flat and then the spherical mirrors after May07. Preparations are underway in 156. We now have an electric crane to lift the boxes onto the rig (operator is needed). Had meetings with Jean-Christophe Gayde to prepare and discuss the survey. Holes for positioning the survey tripods will be drilled next week and Christophe is preparing the optical benches. The tent in 156 will be cleaned before the mirror alignment begins.

By the 4 June07 the beampipe operations will be complete. This is the earliest possible date to install the mirrors.
4. HPD Assembly
Dressing:  Bottom 7 assembled. The cooling plates need to go on and the temperature sensors will be tested (mid May to beginning of June). Then the remaining 9 columns will started. Plan to get all column dressing finished (including spares). The missing cups will be ordered (30 pieces) (Action Gabriel).
Magnetic Shield: Richard P has completed insulation of the remaining RICH1 magnetic shields.  These will be tested (action Barbara).

Upper Rail System:  Expected to be ready the 3rd week of May07.
PPSPS:  Documents in EDMS. Mail sent to Rolf and Erik, waiting for a reply.

Lower Rail System: Brian distributed a concept design. Meeting with Brian will be arranged at IC next week to discuss the details and elaborate the design.

Cabling:  The upper cable tray was installed this week. The cables will be laid during the second half of May07. Maria and Vera will come out to put on the connectors at the start of June 07.
Light Leak Detector:  No news. Plan to use two small PM tubes per box.
5.  MDMS
Ray presented the status of the MDMS at this week’s RICH group meeting. The concept was well received. The system will not cover all the HPD’s. The timescale is optimistic and tight i.e. ready by July07. 
6. LAMS
Three cameras and the box are at CERN and they will be tested (action Richard P.).
7. Aerogel
The trail mounting of the aerogel boxes was done last week with the 300 micron glass window on. The insertion of the aerogel boxes was not complete because of the beampipe protection. The next trail could be done just after the flat mirror installation.
8. Quartz

The company will deliver the frames after 5 weeks from receiving the order. The order is not yet sent, but will be sent soon. The gluing onto the frames can start not before July07.
9. Next meeting
Friday 11th May 2007 at 10:00(GMT) 11:00(CET). 
